Author's response to reviews

Title: Comparison of foot orthoses made by podiatrists, pedorthists and orthotists regarding plantar pressure reduction.

Authors:

Nick A Guldemond (ngu@orthop.unimaas.nl)
Pieter Leffers (p.leffers@epid.unimaas.nl)
Nicolaas C Schaper (nsc@sint.azm.nl)
Antal P Sanders (asan@rev.azm.nl)
Fred HM Nieman (fni@oazg.azm.nl)
Geert HIM Walenkamp (gwa@sort.azm.nl)

Version: 7 Date: 9 November 2005

Author's response to reviews: see over
Dear Dr. Pemberton,

On behalf of our project group, we would like to thank the editorial board and the referees for the helpful comments and the positive decision for publication of the manuscript "Comparison of foot orthoses made by podiatrists, pedorthists and orthotists regarding plantar pressure reduction" in the BMC journal *Musculoskeletal Disorders*.

Yours sincerely,

Nick Guldemond, clinical researcher
On behalf of the fellow authors

Dept. Orthopaedic Surgery
University Hospital Maastricht
P.O. Box 5800
6202 AZ Maastricht
The Netherlands
Phone: +31 (0)43 3875031
Fax: +31 (0)43 3874893
Mobile: +31 (0)6 42393696

e-mail: ngu@orthop.unimaas.nl